Author's response to reviews

Title: Mapping of health system functions to strengthen priority programs. The case of maternal health in Mexico.

Authors:

   Miguel Angel Gonzalez Block (miguel.gonzalezblock@insp.mx)
   Mariel V. Rouver (mariel.rouvier@insp.mx)
   Víctor Becerril (victor.becerril@insp.mx)
   Paola M. Sesia (sesia@ciesas.edu.mx)

Version: 2 Date: 3 November 2010

Author's response to reviews:

Dear BMC Public Health

Thanks very much for the two reviewers’ comments to our paper “Mapping of health system functions to strengthen priority programs. The case of maternal health in Mexico”. We responded to all comments, which were most useful to improve the paper. We have submitted the revised paper and a table where we identify how each comment was addressed.

In addition to the reviewer comments we shortened the title of our paper.

Thanks again for considering the paper for publication.

Best regards